Title: Exposure to cooking fuels and birth weight in Lanzhou, China: a birth cohort study

Authors:

Min Jiang (athene.jiang@hotmail.com)
Jie Qiu (qioujie@21cn.com)
Min Zhou (ghz66383399@sina.com)
Xiaochun He (LZxiaochunhe@126.com)
Hongmei Cui (369678469@qq.com)
Catherine Lerro (catherine.lerro@yale.edu)
Ling Lv (1358638351@qq.com)
Xiaojuan Lin (jxinlinmail@163.com)
Chong Zhang (zhch1972@163.com)
Honghong Zhang (64276995@qq.com)
Ruifeng Xu (xrf730130@163.com)
Daling Zhu (zdlldz@126.com)
Yun Dang (34370100@qq.com)
Xudong Han (hxd809904@126.com)
Hanru Zhang (zhanghrfred@163.com)
Haiya Bai (17993902@qq.com)
Ya Chen (cty704@163.com)
Zhongfeng Tang (tzf3325@gmail.com)
Ru Lin (1632624248@qq.com)
Tingting Yao (yaott821005@163.com)
Jie Su (50249405@qq.com)
Xiaoying Xu (794983177@qq.com)
Xiaohui Liu (lxhpxq609@163.com)
Wendi Wang (xiaodismile@hotmail.com)
Yueyuan Wang (wangyueyuan0618@163.com)
Bin Ma (153873545@qq.com)
Weitao Qiu (bjbqwt@qq.com)
Cairong Zhu (cairong.zhu@hotmail.com)
Suping Wang (spwang88@163.com)
Huang Huang (h.huang@yale.edu)
Nan Zhao (nan.zhao@yale.edu)
Xiaosong Li (lixiaosong1101@126.com)
Qing Liu (2305470816@qq.com)
Yawei Zhang (yawei.zhang@yale.edu)

Version: 4
Date: 3 June 2015

Author's response to reviews: see over
Dear Editor,

The manuscript was read and edited by a native speaker, Catherine Lerro, who is also a coauthor on this paper. We hope that these revisions will now make it possible for the manuscript to be published in your journal.

If you have any further suggestions, please let us know. I can be reached by phone at 203-785-6210; by fax at 203-737-6023, and by e-mail at yawei.zhang@yale.edu.

Thank you and best wishes.

Sincerely yours,

Yawei Zhang, M.D., Ph.D., M.P.H.
Associate Professor
Yale University School of Public Health
60 College Street, LEPH 440
203-7856210 (phone)
203-7376023 (fax)
yawei.zhang@yale.edu (email)